# Treatment of first-time traumatic anterior shoulder dislocation: the UK TASH-D cohort study

Jonathan L Rees, <sup>1,2</sup>\* Anjali Shah, <sup>1,2</sup> Katherine Edwards, <sup>1,2</sup> Maria T Sanchez-Santos, <sup>1,2</sup> Danielle E Robinson, <sup>1,2</sup> Antonella Delmestri, <sup>1,2</sup> Andrew Carr, <sup>1,2</sup> Nigel Arden, <sup>1,2,3</sup> Sarah E Lamb, <sup>1,2,7</sup> Amar Rangan, <sup>1,2,4,5</sup> Andrew Judge, <sup>1,2</sup> Rafael Pinedo-Villanueva, <sup>1,2</sup> Tim Holt, <sup>6</sup> Sally Hopewell, <sup>1,2,7</sup> Daniel Prieto-Alhambra<sup>1,2</sup> and Gary Collins<sup>1,2</sup>

<sup>1</sup>Nuffield Department of Orthopaedics, Rheumatology and Musculoskeletal Sciences, University of Oxford, Oxford, UK

- <sup>2</sup>NIHR Oxford Biomedical Research Centre, Oxford, UK
- <sup>3</sup>MRC Lifecourse Epidemiology Unit, University of Southampton, Southampton, UK <sup>4</sup>Department of Health Sciences, University of York, York, UK

<sup>5</sup>The James Cook University Hospital, South Tees Hospital NHS Foundation Trust, Middlesbrough, UK

<sup>6</sup>Nuffield Department of Primary Care Health Sciences, University of Oxford, Oxford, UK

<sup>7</sup>Oxford Clinical Trials Research Unit, University of Oxford, Oxford, UK

\*Corresponding author jonathan.rees@ndorms.ox.ac.uk

Declared competing interests of authors: Nigel Arden has received honoraria from, held advisory board positions (which involved receipt of fees) in and received consortium research grants from Merck & Co. (Kenilworth, NJ, USA) (honorarium), Roche Holding AG (Basel, Switzerland), Novartis (Basel, Switzerland) and Bioiberica S.A. (Barcelona, Spain) (grants), Smith & Nephew plc (London, UK), NicOx S.A. (Valbonne, France), Flexion Bioventus (Bioventus LLC, Durham, NC, USA) and Freshfields Bruckhaus Deringer LLP (London, UK) (personal fees) outside the submitted work. Amar Rangan reports grants from the National Institute for Health Research (NIHR) Health Technology Assessment (HTA) programme, Orthopaedic Research UK (London, UK), DePuy Synthes UK (Leeds, UK) and JRI Orthopaedics (Sheffield, UK) outside the submitted work. Andrew Judge is a subpanel member of the NIHR Programme Grants for Applied Research (PGfAR) programme, has received consultancy fees from Freshfields Bruckhaus Deringer LLP and has held advisory board positions (which involved receipt of fees) from Anthera Pharmaceuticals Inc. (Hayward, CA, USA). Daniel Prieto Alhambra has received grants and other support from Amgen Inc. (Thousand Oaks, CA, USA) and UCB Biopharmal Srl (Brussels, Belgium); grants from Laboratories Servier (Neuilly-sur-Seine, France), Novartis International AG (Basel, Switzerland), Astellas Pharma Inc. (Tokyo, Japan), the NIHR HTA programme and from NIHR Research for Patient Benefit (RfPB), outside the submitted work. He is also a member of the NIHR HTA Clinical Evaluation and Trials panel (from November 2017 to present) and the NIHR RfPB South-Central Regional Advisory

Committee panel (from 2013 to 2017). Tim Holt is a general practitioner (GP) in London and is a GP advisor for, but not employed by, the Clinical Practice Research Datalink. Gary S Collins is a member of the HTA Commissioning Board and has received grants from the NIHR HTA programme, NIHR RfPB, NIHR Biomedical Research Centre (BRC) and British Heart Foundation outside the submitted work. Sarah E Lamb was on the HTA Additional Capacity Funding Board (2012–15), HTA End of Life Care and Add-on Studies Board (2012–15), HTA Prioritisation Group Board (2010–15), HTA Trauma Board (2013–15), HTA Clinical Trials Board (2010–15) and the HTA Funding Boards Policy Group (2010–15) within 36 months of the start of the study. Andrew Carr has received other grants from the NIHR HTA programme, the Medical Research Council and the Wellcome Trust during the conduct of this study. He is a panel member on the Medical Research Council Developmental Pathway Funding Scheme (2016–present), a theme leader for the NIHR Oxford Biomedical Research Centre (2017–present) and was the Director of the NIHR Oxford Musculoskeletal Biomedical Research Unit (2008–17). Jonathan L Rees has received other grants from the NIHR HTA and NIHR PGfAR programmes. He works within a NIHR BRC and currently holds other grants from the Royal College of Surgeons of England, the Dinwoodie Charitable Company (Macclesfield, UK), McLaren Applied Technologies (Woking, UK) and the National Joint Register (NJR). He sits on committees at the NJR, National Institute for Health and Care Excellence and the Orthopaedic Data Evaluation Panel (ODEP), advises the Medicines and Healthcare Products Regulatory Agency and is a council member of the British Elbow and Shoulder Society.

Published April 2019 DOI: 10.3310/hta23180

# **Plain English summary**

## UK TASH-D cohort study

Health Technology Assessment 2019; Vol. 23: No. 18 DOI: 10.3310/hta23180

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

Traumatic anterior shoulder dislocation (TASD) happens when the top of the arm bone is forced frontwards out of the shoulder socket. After a TASD, the shoulder joint can become 'unstable' and keep dislocating. The main treatments are surgery or physiotherapy; however, we do not know which treatment is best at stopping more dislocations.

Two large NHS computer databases were studied to assess this problem. This has allowed us to produce information on the extent of this problem in the UK. We also looked for any differences in the number of people who suffered more shoulder dislocations when treated with either surgery or no surgery.

The results showed that young men aged 16-20 years and women aged > 50 years suffer the most with this problem. In young people, the cause is thought to be due to sports injuries. These findings in women aged > 50 years are new and suggest that further research is needed to discover what puts them at a greater risk of TASD.

When patients who had surgery and those who did not were compared, there appeared to be no difference in the number of people suffering a re-dislocation. Although, overall, this might suggest that surgery after only one dislocation does not have any extra benefit in preventing more dislocations, this research discovered that important information used to help decide on whether or not surgical treatment is needed is not reported in the databases. Some patients may be at a greater risk of more dislocations than other patients based on risk factors, such as sport and occupation, and this information is not recorded in the NHS databases. Therefore, the research question cannot be answered by studying these NHS databases and so other methods, such as a research trial or a custom database built especially for shoulder dislocation patients, would be needed.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Rees *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

## **Health Technology Assessment**

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.513

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

#### **HTA programme**

The HTA programme, part of the National Institute for Health Research (NIHR), was set up in 1993. It produces high-quality research information on the effectiveness, costs and broader impact of health technologies for those who use, manage and provide care in the NHS. 'Health technologies' are broadly defined as all interventions used to promote health, prevent and treat disease, and improve rehabilitation and long-term care.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

For more information about the HTA programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hta

#### This report

The research reported in this issue of the journal was funded by the HTA programme as project number 14/160/01. The contractual start date was in January 2016. The draft report began editorial review in June 2018 and was accepted for publication in October 2018. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Rees *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

## **NIHR Journals Library Editor-in-Chief**

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

## **NIHR Journals Library Editors**

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Honorary Professor, University of Manchester, and Senior Clinical Researcher and Associate Professor, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

**Dr Catriona McDaid** Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk